Author's response to reviews

Title: Occupation, smoking, and chronic obstructive respiratory disorders: a cross sectional study in an industrial area of Catalonia, Spain

Authors:

Angeles Jaen (aim.ceescat.germanstrias@gencat.net)
Jan-Paul Zock (jpzock@imim.es)
Manolis Kogevinas (kogevinas@imim.es)
Antonio Ferrer (AFERRER@CSPT.ES)
Albert Marin (amarin@cspt.es)

Version: 3 Date: 23 January 2006

Author's response to reviews:

Response to the suggestions for technical revisions

Many thanks for the opportunity to revise the details necessary for acceptance in your journal. We have accepted all suggested changes.

. Title page: The authors' affiliations are as follows:

1Centre d'Estudis Epidemiologics sobre la SIDA de Catalunya (CEESCAT), Crta de Canyet s/n, 08916 Badalona, Spain

2Respiratory and Environmental Health Research Unit, Institut Municipal d'Investigacio Medica (IMIM), Dr. Aiguader 80, 08003 Barcelona, Spain

3Hospital de Sabadell. Consorci Hospitalari Parc Tauli, Parc Tauli s/n 08208 Sabadell; Spain

The zipcode for #2 has been included. Street addresses (please note that offices in Spain often have "s/n" instead of a street number!) are included for all three.

. All authors of publication #20 have been included (instead of "et al").

. Table 2 and table 4 have been included as modified by the technical editor.